Author's response to reviews

Title: [Cost]effectiveness of withdrawal of fall-risk increasing drugs versus conservative treatment in older fallers: design of a multicenter randomized controlled trial (IMPROveFALL-study)

Authors:

Klaas A Hartholt (k.hartholt@erasusmc.nl)
Nathalie Van der Velde (n.vandervelde@erasusmc.nl)
Esther MM Van Lieshout (e.vanlieshout@erasusmc.nl)
Suzanne Polinder (s.polinder@erasusmc.nl)
Oscar J De Vries (oj.devries@vumc.nl)
Nicole DA Boyé (n.boye@erasusmc.nl)
Albert JH Kerver (b.kerver@sfg.nl)
Gijsbertus Ziere (g.ziere@havenziekenhuis.nl)
Milko MM Brujininckx (mbrujininckx@ysl.nl)
Mark R De Vries (vriese@rdgg.nl)
Francesco US Mattace-Raso (f.mattaceraso@erasusmc.nl)
André G Uitterlinden (a.g.uitterlinden@erasusmc.nl)
Ed F Van Beeck (e.vanbeeck@erasusmc.nl)
Paul Lips (p.lips@vumc.nl)
Peter Patka (p.patka@erasusmc.nl)
Tischa JM Van der Cammen (t.vandercammen@erasusmc.nl)

Version: 5  Date: 16 August 2011

Author's response to reviews:

Dear Production Team,

Thank you for your e-mail with the questions for the formatting changes.

1. The figure must be cited in the manuscript: done
2. Abstract: Please change conclusion to discussion: done (submission system and manuscript)

If you have any other questions, please don't hesitate to contact us.

Best regards,

Klaas Hartholt and Tischa van der Cammen